Author's response to reviews

Title: Acute partial Budd-Chiari syndrome and portal vein thrombosis in cytomegalovirus primary infection

Authors:

Laurent Spahr (Laurent.Spahr@hcuge.ch)
Andreas Cerny (andreas.cerny@bluewin.ch)
Isabelle Morard (Isabelle.Morard@hcuge.ch)
Laura Rubbia-Brandt (rubbia-brandt-laura@diogenes.hcuge.ch)
Jacques Schrenzel (jacques.schrenzel@hcuge.ch)

Version: 2 Date: 24 October 2005

Author's response to reviews:

Dear Editor,
I have made the changes that your required. The following changes have been made:
1. title: ...a case report (added)
2. Abstract: (166 words). The abstract has been rewritten and structured (Backgr'nd; Case Pres; Conclusions)
3. Manuscript: some details have been added in the text